Author’s response to reviews

Title: Point-of-Care Measurement of Activated Clotting Time for Cardiac Surgery as measured by the Hemochron Signature Elite and the Abbott i-STAT: Agreement, Concordance, and Clinical Reliability

Authors:

Daniel Dirkmann (Daniel.Dirkmann@uk-essen.de)
Elisabeth Nagy (Elisabeth.Nagy@uk-essen.de)
Martin Britten (martin.britten@uk-essen.de)
Juergen Peters (juergen.peters@uni-duisburg-essen.de)

Version: 2 Date: 27 Aug 2019

Author’s response to reviews:

Response to the Editor: Thank you for your time and effort spent.

1. As requested, we have added 'by the ethics committee' in order to clarify who waived the need for consent.

2. Figure 2 is now mentioned in the manuscript.

3. The wording regarding funding has been changed, as requested, now reading 'No funding was obtained for this study.'

4. As requested, only files that shall be published are uploaded.

5. All files are clean versions.

Response to reviewer 2: Thank you for again spending time and effort in reviewing our manuscript. The typo on page 8, line 8 has been corrected.